Author's response to reviews

Title: Three-dimensional quantitative analysis of adhesive remnants and enamel loss resulting from debonding orthodontic molar tubes

Authors:

Joanna Janiszewska-Olszowska (jjo@pum.edu.pl)
Katarzyna Tandecka (katarzyna.tandecka@tu.koszalin.pl)
Tomasz Szatkiewicz (tomasz.szatkiewicz@tu.koszalin.pl)
Katarzyna Sporniak-Tutak (katarzyna@aestheticdent.pl)
Katarzyna Grocholewicz (kgrocholewicz@o2.pl)

Version: 5 Date: 31 August 2014

Author's response to reviews:

Dear Sirs,

The authors are very grateful for the detailed revision of the manuscript. We have addressed every of the three minor corrections by the Reviewer Professor Abbas R Zaher.

1. In Material and Methods Section, 3rd Paragraph, 4th line: “... at the power of the test of 80 ... “
   The number “80” should be changed to “0.80”
   This has been corrected as proposed.
2. In References Section, 10th Reference: “Srp Arh Celok Lek2012”
   A space is missing between “Lek” and “2012”
   The authors have checked, that no space was missing, but the title is in Italic according to the Instruction for authors, thus it looked as there were no space.
3. In References Section, 11th Reference:
   This article has a very strange page numbering from 387.e1 to 387.e9 and is followed by a discussion on pages numbered from 387 to 389. Thus it seems, that this citation is correct. This article is cited in PubMed in the same manner.